Author's response to reviews

Title: Quantifying the reduction in general practitioner referral rates from participation in joint teleconsultations: a discrete choice experiment.

Authors:

Tiago Cravo Oliveira (t.cravo-oliveira09@alumni.imperial.ac.uk)
James Barlow (j.barlow@imperial.ac.uk)
Steffen Bayer (steffen.bayer@duke-nus.edu.sg)

Version: 2 Date: 16 September 2014

Author's response to reviews:

Thank you for your response. In light of the editorial requests made, we have:
1) Added a statement on ethical approval to the end of the Methods section.
2) Included line numbering and removed page breaks.
3) Added a section on the "Availability of supporting data".
4) Included a section with a "List of abbreviations".
5) Removed Figure 1 from the manuscript while retaining the legend.
We hope these changes are to your satisfaction. We look forward to receiving your feedback.
Yours sincerely,
Tiago Cravo Oliveira